Author's response to reviews

Title: One stop or full stop? The continuing challenges for researchers despite the new streamlined NHS research governance process

Authors:

Andrew G.H. Thompson (Andrew.Thompson@ed.ac.uk)
Emma F France (Emma.France@stir.ac.uk)

Version: 3 Date: 30 April 2010

Author's response to reviews:

We have carried out the following changes:

- Amended the title of Figure 2. It previously read: 'Figure 2 NHS Governance Process Approvals and Permission (excluding non-responses).' It now reads: 'Figure 2 Time taken for NHS Governance Approvals and Permission (excluding non-responses).'</n
- Added missing reference 24.
- Amended Figure 1 and Figure 2 to cut out white border and to enhance readability of Figure 1 by some small changes to format of boxes, the key and to the 'time' arrow which is now uni-directional.
- Renamed the glossary section 'Abbreviations', and re-formatted into one paragraph, as requested.